Author's response to reviews

Title: The United States and Canada as a coupled epidemiological system: an example from hepatitis A

Authors:

Raluca Amariei (ramariei@uoguelph.ca)
Allan Willms (awillms@uoguelph.ca)
Chris T Bauch (cbauch@uoguelph.ca)

Version: 7 Date: 27 February 2008

Author's response to reviews:

Please move the appendices to directly before the acknowledgments.

THE APPENDICES HAVE BEEN MOVED.

References 15, 33, 34, and 35 are cited in the table at the end of the manuscript. Although every effort will be made to place the tables within the text at the appropriate place, please cite the references in the table as if the tables are at the end of the manuscript. Please reorder your references accordingly,

REFERENCE 15 SHOULD HAVE BEEN CITED IN THE INTRODUCTION JUST AFTER REFERENCE 14 (THE CITATION WAS SOMEHOW DROPPED). THE CITATION TO REFERENCE 15 HAS BEEN PUT BACK INTO THE INTRODUCTION, AFTER THE CITATION TO REFERENCE 14, SO IT IS NOT NECESSARY TO RENUMBER REFERENCES 15-33

REFERENCES 33-36 HAVE BEEN MOVED TO THE BACK OF THE REFERENCE LIST, AND ALL INTERVENING REFERENCES HAVE BEEN SHIFTED UP BY THREE. CORRESPONDING CITATIONS HAVE BEEN CHANGED.

We also notice that reference 21 is mentioned before reference 20. Please ensure that you correct this.

THIS HAS ALSO BEEN CORRECTED.

THANKS,

DR. BAUCH